Author’s response to reviews

**Title:** The Relation Between Neutrophil-to-Lymphocyte Ratio and Coronary Chronic Total Occlusions

**Authors:**

Kenan Demir (drkenan76@yahoo.com)
Ahmet Avci (drahmetavci@yahoo.com)
Bulent B Altunkeser (bbaltunkeser@hotmail.com)
Ahmet Yilmaz (dr.ahmetyilmaz@gmail.com)
Fikret Keles (keles.fikret@gmail.com)
Ahmet Ersecgin (ersecgin@gmail.com)

**Version:** 8  **Date:** 11 September 2014

Author’s response to reviews: see over
Dear Editor,

10 July 2014

We are submitting a manuscript (original article) entitled “The Relation Between Neutrophil-to-Lymphocyte Ratio and Coronary Chronic Total Occlusions”. All authors state that this manuscript, or portions thereof is not under consideration by another journal and has not been previously published.

We state that everyone who performed this study is included in the author list and the authors are ranked according to their contribution. All authors have read and approved the manuscript. There is no conflict of interest exists.

Thank you for your kind consideration for publication.

Yours Sincerely

Correspondence author: Kenan Demir
Mailing address: Selcuk University Faculty of Medicine
Cardiology Department, 42075
Konya / TURKEY
Telephone number: +90 505 5061334
Fax number: +90 332 241 2151
E-mail: drkenan76@yahoo.com